Dear Editor,

Thank you for carefully reviewing our manuscript previously titled “Programmed cell death 1 (PD-1) antibody in combination with chemotherapy for treatment of SMARCA4-deficient advanced undifferentiated carcinoma of the duodenum: case report ” for possible publication in the World Journal of Clinical Oncology. We are grateful to you and your reviewers for their constructive critique. We have revised the manuscript, highlighting our revisions in yellow, and have attached point-by-point responses detailing how we have revised the manuscript in response to the reviewers’ comments below.

Thank you for your consideration and further review of our manuscript. Please do not hesitate to contact us with any further questions or recommendations.

Yours Sincerely,

Yi-Fan Li, Wenhui Yang

Reviewer Comments:

1. A timeline summarizing the main events of this case report should be included. It would help the readability of the paper

   Response: done. Timeline summarizing was presented in Figure 3

2. The discussion should be expanded and a more personal perspective should be included - The background of the role of cancer IO should be better discussed, and some recent papers regarding this topic should be included

   Response: done. The revised content was highlight in yellow as reviewer's-instruction.

3. Language polishing requirements for revised manuscripts was needed.

   Response: done. New language certificate along with the manuscript were presented.